Author's response to reviews

Title: Utilization of PMTCT services and associated factors among pregnant women attending antenatal clinics in Addis Ababa, Ethiopia

Authors:

Wakgari Deressa (deressaw@yahoo.com)
Assefa Seme (assefaseme@gmail.com)
Anteneh Assefa (antex98@yahoo.com)
Getachew Teshome (getchkid@yahoo.com)
Fikre Enqusellassie (fikreens@yahoo.com)

Version: 2 Date: 25 March 2014

Author's response to reviews: see over
Dear Jane,

Thank you for sending us your prompt comments on our manuscript MS: 7114311651249285 [Deressa et al. Utilization of PMTCT services and associated factors among pregnant women attending antenatal clinics in Addis Ababa, Ethiopia].

As per your suggestions, we have inserted line and page numbers on the revised MS. In addition, we have also removed any page numbers from the revised MS.

We hope that the revised manuscript now addresses all of your. Please find attached the point-by-point response to these comments as well as the revised manuscript.

If you have any further queries, please do not hesitate to contact us at the contact addresses given below.

We look forward to hearing from you soon.

Kind regards,

Wakgari Deressa, PhD.
Corresponding author
School of Public Health, Addis Ababa University
P. O. Box 6850
E-mail: deressaw@yahoo.com; deressaw@gmail.com
Addis Ababa, Ethiopia.